Author’s response to reviews

Title: Qualitative exploration of the acceptability of a postnatal pelvic floor muscle training intervention to prevent urinary incontinence

Authors:

Aileen Grant (a.grant17@rgu.ac.uk)

Sinead Currie (Sinead.currie@stir.ac.uk)

Version: 2 Date: 11 Dec 2019

Author’s response to reviews:

Qualitative exploration of the acceptability of a postnatal pelvic floor muscle training intervention to prevent urinary incontinence.

Improvements to the English language of the manuscript have been requested. The manuscript has been reviewed by someone fluent in English and appropriate revisions made.

For the 'availability of data and materials' section justification for why the data will not be made publicly available has been given. The authors did not obtain informed consent to make all the data publicly available and only have informed consent for the use of the data in academic publications. We are happy to make the data available to those who have a reasonable request.